communications medicine

ARTICLE

https://doi.org/10.1038/s43856-023-00246-4

OPEN

Heath management app use in Parkinson's disease and quality of life during the COVID-19 pandemic

Yilin Tang ^{1,4}, Xiaoniu Liang^{1,4}, Bo Shen^{1,4}, Jiawei Wang², Bastiaan R. Bloem³, Jianjun Wu ¹[∞] & Jian Wang ¹[∞]

Abstract

Background Social distancing during the COVID-19 pandemic affected follow-up visits and medication availability for patients with Parkinson's disease (PD). As a promising strategy to deal with these challenges, the implementation of health management smartphone apps was accelerated. However, whether more intense use of such apps could improve the quality of life (QoL) for PD patients during the COVID-19 pandemic was unknown.

Methods Using a PD management app, this observational study assessed changes in QoL, as determined by PD Questionnaire 8 (PDQ-8), among PD patients before (Jan 20, 2019 to Oct 6, 2019) and after the beginning of the COVID-19 lockdown (Jan 20, 2020 to Oct 6, 2020). According to adherence to use of the app, participants were divided into low adherence, moderate adherence, and high adherence groups. A total of 4979 PD patients registered in the app, and 226 PD patients were enrolled, including 57 patients with low adherence, 112 with moderate adherence and 57 with high adherence. A generalized linear model was used to evaluate the change of PDQ-8 scores across these three different adherence groups.

Results After the COVID-19 lockdown (1-year follow-up), the PDQ-8 scores are reduced by 0.8 (95% Cl, 0.3-1.4) in all participants (P = 0.004). After adjustment for age, gender, education, disease duration and levodopa equivalent dose, PDQ-8 scores significantly less reduced in the high adherence group (0.3; 95% Cl, 0.6-1.2) compared to the low adherence (1.9; 95% Cl, 0.7-3.1) (P = 0.040) and moderate adherence groups (0.6; 95% Cl, 0.2-1.3) (P = 0.012).

Conclusions A health management smartphone-based app might be a way to both measure and improve QoL among PD patients, provided that sufficient adherence is achieved.

Plain language summary

The COVID-19 pandemic had an enormous impact on the lives of people living with Parkinson's disease (PD), given social distancing measures, and reduced access to healthcare. As a way to mitigate this, the use of health management apps was accelerated. However, there is a lack of studies on whether the use of such apps could over time affect the quality of life of people living with PD. Here, we analyzed changes in the quality of life of people living with PD using a health management app, before and after social distancing. We found that patients with high adherence to use of the app experienced a lower reduction of their quality of life. Using a health management smartphone app represents a novel approach that might help patients with PD improve their quality of life.



¹Department of Neurology and National Research Center for Aging and Medicine & National Center for Neurological Disorders, State Key Laboratory of Medical Neurobiology, Huashan Hospital, Fudan University, Shanghai, China. ² College of Arts and Science, Vanderbilt University, Nashville, TN 37212, USA. ³Department of Neurology, Radboud University Medical Centre, Donders Institute for Brain, Cognition and Behaviour, Centre of Expertise for Parkinson and Movement Disorders, Nijmegen, The Netherlands. ⁴These authors contributed equally: Yilin Tang, Xiaoniu Liang, Bo Shen. ¹²email: wujianjun@fudan.edu.cn; wangjian_hs@fudan.edu.cn

he ongoing COVID-19 pandemic has an enormous impact on the lives of many people worldwide. Some national governments have introduced extensive 'lockdown' measures including social distancing to contain the spread of transmission of COVID-19. Individuals with a chronic disease, including Parkinson's disease (PD), are particularly vulnerable to such restrictions. There is limited capacity for follow-up visits, a restricted medication availability, less options to engage in regular physical activity, including inability to participate in exercise classes, and increased levels of psychological distress^{1,2}.

Mobile health interventions, such as videoconferencing or smartphone applications (apps), have been advocated as promising strategies to assist in the management of PD³. These telemedicine tools have the potential to ascertain routine follow-up visits remotely, organize new subspecialty consultations, perform remote research visits, and also to deliver interventions remotely, including psychotherapy, social services, rehabilitation, and education. The merits of telemedicine interventions are supported by a small but growing body of evidence^{4–10}. Potential benefits include cost savings, a reduction in travel distance, and a reduction in time spent attending appointments. However, most studies using telemedicine tools for patients with PD were developed only for specific self-management properties such as medication diary, remote consultation, or exercise.

Currently, the COVID-19 crisis is catalyzing the use of mobile health interventions, which have now become an attractive alternative route of delivering care to PD patients while adhering to the social distancing measures^{11,12}. To our knowledge, no study has tested whether the use of telemedicine strategies could affect a clinically important outcome—namely quality of life (QoL)—in PD patients during the COVID-19 pandemic. In this observational study, we first assess the change in QoL for PD patients before and after the COVID-19 lockdown using a novel smartphone-based app, that is designed to remotely cover most of the aspects regarding the management of PD¹³. Then, we seek to test whether more intense use of the app would be associated with better QoL during the COVID-19 pandemic.

Methods

Study Population. PD was diagnosed by a specialist in movement disorders according to the Movement Disorder Society Clinical Diagnostic Criteria during an in-clinic evaluation¹⁴. PD patients who owned IOS or Android smartphones were invited to download an app developed by our group as a not-for-profit effort to support the management of PD remotely (Pawei-which is Chinese for "for Parkinson's disease"). Details about the app are provided below. Participants were provided with instructions on how to use the app. After completing the registration, the app directs the participants to an informed consent page, where the participants can either "agree" or "decline" their participation. The study was approved by the Institutional Review Board of the Huashan Hospital (approval number: KY2016-214). The consents are provided by all patients, but given the remote nature of the application does not require a physical signature. The study is registered at the ClinicalTrials.gov database under registration number NCT03649503. Since social distancing began officially in China on Jan 20, 2020, inclusion criteria included, (i) a diagnoses of idiopathic PD, (ii) access to a smartphone on a daily basis at home, (iii) completed both a self-evaluation prior to introduction of these measures (baseline scores, between Jan 20, 2019 and Oct 6, 2019) and a follow-up self-evaluation one year later (between Jan 20, 2020 and Oct 6, 2020) (detailed workflow in Supplementary Fig. 1). Key exclusion criteria included, (i) suspected Parkinsonism due to causes other than idiopathic PD, (ii) a diagnosis of dementia, stroke, epilepsy, encephalitis, traumatic

brain injury, malignancies, or severe psychiatric illness (as recorded in patient case file).

Following consent, participants are asked to complete selfevaluation questionnaires, at baseline and at least once each season (at 90-day intervals) for regular evaluation. At each evaluation, participants completed the Parkinson Disease Questionnaire 8 (PDQ-8, the original score ranges from 0 to 32)^{15,16}, Geriatric Depression Scale 15 (GDS-15)¹⁷, and Movement Disorder Society-Sponsored Revision of the Unified Parkinson's Disease Rating Scale (MDS-UPDRS) IB and II¹⁸. Furthermore, patients were instructed to complete information about previous and current medication use. The dosage of anti-parkinsonian drugs was converted into a total daily LED¹⁹.

The PD remote management app. In April 2017, we launched a platform for iOS and Android smartphone applications for Chinese patients and doctors¹³. At the time of the start of this present study (Jan 20, 2019), 126 hospitals across China had joined this platform.

The app consisted of the following features (Supplementary Fig. 2):

- 1. Episodic delivery of various self-evaluation questionnaires;
- 2. Establishing remote communication between patients and physicians, including post-intervention feedback;
- 3. Offering educational material about PD management (articles, audios and videos) and illustrative educational case histories;
- 4. Evaluating disease severity based on uploaded video and voice recordings (self-recorded by patients at home).

Adherence to use the app. We quantified the frequency of using the app based on three elements: (1) the number of times per month when patients accessed the educational materials; (2) the number of remote patient-physician consultations; and (3) the number of completed episodic self-evaluation questionnaires. The raw number of those times was transformed into a z-score by subtracting the mean number of times from the participant, and then dividing the difference by the standard deviation of the number of times according to the following formula:

Z-score = (number of times – mean)/SD.

The mean of three z-scores for each part was used as the compound score of app usage frequency. Using the distribution of the compound score, the following definitions were created: low adherence: compound score below the 25th percentile; moderate adherence: compound score \geq 25th percentile and \leq 75th percentile; and high adherence: compound score greater than 75th percentile.

Outcomes. The primary outcome was change in QoL, as determined by PDQ-8 before and after the COVID-19 lockdown.

Statistical analysis. Categorical variables were expressed as frequencies (%), and continuous variables were expressed as the mean \pm standard deviation (SD). Across the three groups (low adherence, moderate adherence, and high adherence), the Chisquared test was used for comparing the categorical variables, and the Kruskal–Wallis test or one-way ANOVA for comparing continuous variables. Wilcoxon signed-rank test was used for two paired groups. The generalized linear model (GLM) was used to assess the association between PDQ-8, GDS-15, MDS-UPDRS IB and II scores and different adherence group among the participants. Unadjusted models were not adjusted for any covariate, and adjusted models were controlled for age, gender, education, disease duration, and LED at baseline. Spearman rank correlation

Table 1 Baseline and 1-year follow-up clinical features of participants.

Variables	Total (<i>N</i> = 226)	Low adherence (N = 57)	Moderate adherence (N = 112)	High adherence (<i>N</i> = 57)	P value ^a
No. of reading the educational materials,	9.1 (15.9)	4.8 (4.4)	7.4 (8.0)	14.0 (24.4)	0.002
mean (SD)					
No. of self-evaluation, mean (SD)	17.7 (12.2)	11.5 (8.6)	15.1 (8.1)	29.2 (14.4)	<0.001
No. of consultations, mean (SD)	7.0 (4.0)	3.5 (1.5)	6.5 (2.0)	11.0 (3.4)	<0.001
Z-score of app usage frequency, mean (SD)	0.0 (0.5)	-0.5 (0.2)	-0.1 (0.1)	0.7 (0.5)	<0.001
Age, year, mean (SD)	60.2 (11.3)	60.7 (12.2)	60.2 (11.8)	59.4 (9.4)	0.829 ^b
Age at onset, year, mean (SD)	56.4 (11.2)	57.6 (12.1)	56.0 (11.6)	56.0 (9.4)	0.655 ^b
Disease duration, year, mean (SD)	2.6 (3.2)	2.3 (3.0)	3.0 (3.5)	2.2 (2.9)	0.081
BMI, kg/m ² , mean (SD)	24.5 (5.7)	24.9 (8.0)	24.2 (5.1)	24.6 (4.8)	0.711
Education, year, mean (SD)	12.2 (3.9)	13.1 (3.1)	11.4 (4.1)	12.9 (4.0)	0.139 ^b
Gender, male, n (%)	129 (57.1)	29 (50.9)	69 (61.6)	31 (54.4)	0.368 ^c
Custom Hand, right hand, n (%)	33 (82.5)	8 (100.0)	15 (71.4)	10 (90.9)	0.134 ^c
Baseline self-evaluation					
PDQ-8, score, mean (SD)	9.6 (7.2)	9.4 (6.3)	10.3 (7.9)	8.3 (6.4)	0.370
GDS-15, score, mean (SD)	7.1 (2.6)	7.4 (3.1)	7.5 (2.6)	6.5 (2.1)	0.263
MDS-UPDRS IB, score, mean (SD)	7.1 (5.1)	7.3 (4.8)	7.7 (5.4)	6.3 (4.8)	0.413
MDS-UPDRS II, score, mean (SD)	11.6 (7.0)	11.6 (6.0)	11.9 (7.2)	11.4 (7.6)	0.843
LED, mg/day, mean (SD)	440.6 (325.4)	391.6 (272.7)	449.5 (316.3)	472.8 (387.1)	0.384
1-year follow-up self-evaluation					
PDQ-8, score, mean (SD)	10.4 (7.4)	11.3 (7.3)	10.9 (7.8)	8.7 (6.6)	0.096
GDS-15, score, mean (SD)	7.1 (2.6)	8.6 (2.5)	7.0 (2.7)	6.4 (2.3)	0.054
MDS-UPDRS IB, score, mean (SD)	9.3 (6.6)	12.8 (7.7)	9.8 (6.8)	6.3 (4.3)	0.095
MDS-UPDRS II, score, mean (SD)	13.9 (10.4)	19.3 (9.8)	15.0 (11.4)	8.7 (6.2)	0.022
LED, mg/day, mean (SD)	626.6 (340.8)	576.2 (265.7)	613.7 (331.5)	702.0 (411.8)	0.128

BMI body mass index, PDQ-8 Parkinson Disease Questionnaire 8, GDS-15 Geriatric Depression Scale 15, MDS-UPDRS Movement Disorder Society-Sponsored Revision of the Unified Parkinson's Disease Rating Scale, LED levodopa equivalent dose.

^aComparison among the three groups with low adherence, moderate adherence, and high adherence

^bThe continuous variables were compared among the three groups by one-way ANOVA test. The other continuous variables were compared among the three groups by Kruskal-Wallis test. ^cThe Chi-squared test was used for comparing the categorical variables.

was used to assess the correlation between the compound score of app usage frequency and baseline and follow-up information. Two-tailed p values are presented. Differences were considered statistically significant at P < 0.05. The data analysis was conducted by SAS 9.4 (SAS Institute Inc., Cary, NC, USA).

Reporting summary. Further information on research design is available in the Nature Research Reporting Summary linked to this article.

Results

Participants. Since its inception, a total of 4979 PD patients registered at the app and agreed to share their data. 2446 PD patients completed at least one regular self-evaluation. 226 PD patients who completed baseline self-evaluation before social distancing, and the 1-year follow-up self-evaluation were enrolled in this study (Supplementary Table 1).

Baseline characteristics. Clinical characteristics are described in Table 1.

Based on the compound score of app usage frequency, 57 patients were classified as having manifested a low adherence, 112 patients as moderate adherence, and 57 patients as high adherence. The clinical characteristics were well balanced among groups at baseline.

We used spearman rank correlation to assess the correlation between the compound score of app usage frequency and baseline characteristics. As presented in Supplementary Table 2, we found a significant but modest correlation between the compound score and baseline LED (r = +0.14, P = 0.047). We observed no correlation between the compound score and other baseline characteristics. **Quality of life**. At baseline, there was no significant difference in PDQ-8 scores across the three groups (P = 0.370) (Table 1).

After the COVID-19 lockdown (1-year follow-up), the mean (SD) PDQ-8 score increased by 0.8 (4.1) (P = 0.004) in all the participants. The mean (SD) PDQ-8 score increased by 1.9 (4.6) (P = 0.004) in the low adherence group, 0.6 (4.0) (P = 0.137) in the moderate adherence group, and 0.3 (3.5) (P = 0.833) in the high adherence group. This change in PDQ-8 score over time differed significantly across the three groups (Table 2). After adjustment for age, gender, education, disease duration, and LED, the increase in PDQ-8 scores in the high adherence group was less than the low adherence group (P = 0.040) and the moderate adherence group (P = 0.012) (Table 2).

Other non-motor self-evaluations. At baseline, a comparison of GDS-15 score, MDS-UPDRS IB and II score showed no significant difference (Table 1). After adjustment for age, gender, education, disease duration and LED, there was no significant difference in change of MDS-UPDRS IB and MDS-UPDRS II score across the three groups. However, the increase in GDS-15 score in the high adherence group was less than the low adherence group (P = 0.039) (Table 2).

Discussion

Our study showed an expected worsening in QoL (reflected by higher PDQ-8 scores) in PD patients after the COVID-19 lockdown. Additionally, we show that PD patients with high adherence to use of the PD management app experienced less worsening of their QoL compared to those with moderate and low adherence.

The ongoing COVID-19 pandemic has an enormous impact on PD patients. Social distancing has affected follow-up visits and

Table 2 Change i	n PDQ-8, GDS-15,	MDS-UPDRS IB a	and II scores by adh	lerence group.							
Variables	Total (N = 226)	Low adherence	Moderate adherence	High adherence	P value ^a	<i>P</i> value (un: Estimate) ^c	adjusted effec	Ħ	P value (ad	justed effect e	stimate) ^c
		(N = 57)	(N = 112)	(N = 57)		L vs M	L vs H	M vs H	L vs M	L vs H	M vs H
PDQ-8 change	0.8 (4.1)	1.9 (4.6)	0.6 (4.0)	0.3 (3.5)	0.004	0.015*	0.012*	0.414	0.994	0.040*	0.012*
score, mean (SD)											
GDS-15 change	0.3 (1.9)	1.3 (2.7)	0.2 (2.0)	0.1 (1.6)	0.596 ^b	0.667	0.529	1.000	0.180	0.039*	0.349
score, mean (SD)											
MDS-UPDRS IB	1.3 (3.7)	2.0 (3.5)	0.8 (3.7)	1.7 (3.9)	0.906	0.926	0.893	0.989	0.524	0.823	0.392
change score, mean (SD)											
MDS-UPDRS II	1.0 (5.3)	2.7 (7.4)	1.7 (5.6)	-0.1 (4.6)	0.738	0.986	0.945	0.714	0.780	0.553	0.050
change score, mean (SD)											
PDO-8 Parkinson Disease	Duestionnaire 8, GDS-15 Geri.	atric Depression Scale 15.	MDS-UPDRS Movement Disord	ler Society-Sponsored Revisio	on of the Unified Par	kinson's Disease Ra	ting Scale, L Low A	Adherence, M Mode	erate Adherence, H	High Adherence.	
$^{*}P < 0.05$.	ree oronns with low adheren	ure moderate adherence a	and high adherence	-				×	×)	
^b The continuous variables ^c The generalized linear mc ^c Value for unadjusted efit	were compared among the t del was used to assess the <i>a</i> set estimate was the result th	hree groups by one-way A second according to the process of the pr	NDV Test. The other continues, GDS-15, MDS-UPDRS IB and of for any covariates; <i>P</i> Value	ous variables were compared I II scores, and different adhe for adjusted effect estimate	l among the three gr rence group. was the result that th	oups by Kruskal-W he models controlle	allis test. d for age, gender,	education, disease o	duration, and levodo	opa equivalent dose	

ARTICLE

COMMUNICATIONS MEDICINE | https://doi.org/10.1038/s43856-023-00246-4

medication availability. The other highly disconcerting consequences of the pandemic include increased levels of stress and a marked reduced physical exercise. This is associated with worsening of both motor and non-motor symptoms^{2,20-24}. QoL is a crucial outcome indicator, and being able to maintain or even improve QoL is an important objective of treatment and care in PD patients. A previous study investigated the impact of the COVID-19 pandemic on QoL of PD patients. The findings suggested that PD patients had significantly worse PDQ-39 dimensions compared with controls¹. However, no prior study has compared the QoL in PD directly before and after COVID-19 outbreak through smartphone-based app. In our study, we showed that the PDQ-8 scores significantly worsened after the COVID-19 lockdown, indicating that QoL became more jeopardized. However, we cannot exclude the possibility that the worsening of QoL resulted primarily from progression of the underlying disease.

This COVID-19 pandemic calls for the introduction of better self-management strategies that can help patients to better deal with the challenges of social distancing and the other consequences of this crisis. The implementation of telemedicine solutions, including apps designed for the smartphone, was accelerated during this crisis^{11,12}. Whether more intense use of smartphone apps could minimize the possible negative impact of COVID-19 pandemic was hitherto unknown. In this study, we explored the association between adherence of using a selfdeveloped PD management app and QoL among PD patients. Our results demonstrate that PD patients with high adherence (and who therefore utilized more services that were delivered via the app) exhibited less worsening of QoL than the moderate and low adherence group, an effect that persisted after correction for confounding variables. We hypothesize several reasons why a smartphone-based remote management app might improve QoL for PD patients. First, being grounded at home induces or aggravates neuropsychiatric symptoms such as depression^{2,23,25}. The smartphone-based app may have helped to diminish the feeling of social isolation and alleviate depression by providing remote support for people living with PD. An argument in favor of this explanation is the fact that PD patients with high adherence exhibited less worsening of the GDS-15 score than the low adherence group in this study. However, since the mean change in the GDS-15 score between low and high adherence group were modest, whether use of apps have an impact on depression warrant further investigation. Second, the app may have promoted patient self-management, enhanced treatment adherence, and improved the quality of the clinical consultations. In this way, patients may have become active co-actors in the management of their disease, gaining better and more personalized treatment, which is particularly important in a heterogeneous disorder like PD. Third, the educational material about PD management and best practice examples on how to facilitate daily life may have further helped to support patients in their self-management, and also facilitated the remote communication with physicians by minimizing the knowledge gap between doctors and patients. We suspect that the improvement in QoL might have resulted from a mix of these various services, rather than from a single functionality provided by app. However, our observational studies can only reveal association between smartphone application use and QoL, a designed experiment is needed to establish a causal inference.

It is also important to address the fact that many patients did not adhere faithfully to using the app. Out of a total of 4979 registered PD patients, only 49.1% patients completed at least one regular self-evaluation. Several other studies also reported a high dropout rate amongst users of smartphone apps: 26% of apps are used only once and 74% of apps are not used more than 10 times²⁶. Methods to increase participants' engagement with smartphone applications should be an important topic in future mobile health studies. Strategies to improve retention may include developing a simpler, more user-friendly application, providing patients with instant feedback, adding an appropriate incentive into the application, such as the ability to conveniently book clinical appointments, or possibly even financial incentives. Finally, use of briefer questionnaires may be helpful to improve compliance. The extensive 39-item Parkinson's disease Questionnaire (PDQ-39) is the most widely used PD-specific healthrelated QoL questionnaire. To provide greater convenience for patients, the PDQ-8 could be used, which is a briefer version of the PDQ-39 and which can assess QoL more quickly, with less burden for the respondents¹⁵. The validity and reliability of the PDQ-8 have been demonstrated in several studies^{16,27}.

This study has several limitations. First, participants were people who owned IOS or Android smartphones and thus were not representative of the broader PD population. Further studies with PD patients without own smartphones remain needed. Second, we corrected for various potential confounding factors impacting on QoL, including age, gender, education, disease duration, and LED. Nonetheless, we cannot fully exclude a possible influence of other factors that were not evaluated by this app, such as UPDRS-III scores, falls, the existence of caregivers, which might also impact on QoL. Obtaining information about these potentially important factors will be necessary in future studies to minimize potential biases. Finally, the present observation lasted for one year; although this is reasonable for a first orienting studies such as ours, more work remains needed to monitor the effect of a smartphone-based remote management app in the long term. It will also be interesting to examine whether any observed benefits persist after participants stop using the app.

Conclusion

In this study, we found that PD patients with high adherence to use of the PD management app experienced less worsening of their QoL. Our results suggest that a smartphone-based app might be a novel way to both measure and improve QoL among PD patients during the COVID-19 pandemic. Further work in a larger sample size and longer periods of time is recommended to determine whether the benefits of the smartphone app are maintained.

Data availability

The data are available from Huashan Hospital, but restrictions apply to protect the privacy of individuals, so the data are not publicly available. Data are, however, available from the authors upon reasonable request and with permission from Huashan Hospital.

Received: 1 May 2022; Accepted: 23 January 2023; Published online: 11 February 2023

References

- Shalash, A. et al. Mental health, physical activity, and quality of life in Parkinson's disease during COVID-19 pandemic. *Mov. Disord.* 35, 1097–1099 (2020).
- van der Heide, A., Meinders, M. J., Bloem, B. R. & Helmich, R. C. The impact of the COVID-19 pandemic on psychological distress, physical activity, and symptom severity in Parkinson's disease. *J. Parkinsons Dis.* 10, 1355–1364 (2020).
- Espay, A. J. et al. Technology in Parkinson's disease: challenges and opportunities. *Mov. Disord.* 31, 1272–1282 (2016).
- Dorsey, E. R. et al. Randomized controlled clinical trial of "virtual house calls" for Parkinson disease. *JAMA Neurol.* 70, 565–570 (2013).

- Qiang, J. K. & Marras, C. Telemedicine in Parkinson's disease: a patient perspective at a tertiary care centre. *Parkinsonism Relat. Disord.* 21, 525–528 (2015).
- 6. Lakshminarayana, R. et al. Using a smartphone-based self-management platform to support medication adherence and clinical consultation in Parkinson's disease. *NPJ Parkinsons Dis.* **3**, 2 (2017).
- Dorsey, E. R. et al. Increasing access to specialty care: a pilot, randomized controlled trial of telemedicine for Parkinson's disease. *Mov. Disord.* 25, 1652–1659 (2010).
- van Uem, J. M. et al. Twelve-week sensor assessment in Parkinson's disease: Impact on quality of life. *Mov. Disord.* 31, 1337–1338 (2016).
- Wagner, L. et al. Evaluation of an individualized, tablet-based physiotherapy training programme for patients with Parkinson's disease: the ParkProTrain study, a quasi-randomised controlled trial. *BMC Neurol.* 22, 176 (2022).
- Park, Y. et al. Effect of mobile health intervention for self-management on self-efficacy, motor and non-motor symptoms, self-management, and quality of life in people with Parkinson's disease: randomized controlled trial. *Geriatr. Nurs.* 46, 90–97 (2022).
- Bloem, B. R., Dorsey, E. R. & Okun, M. S. The coronavirus disease 2019 crisis as catalyst for telemedicine for chronic neurological disorders. *JAMA Neurol.* 77, 927–928 (2020).
- Dorsey, E. R., Bloem, B. R. & Okun, M. S. A new day: the role of telemedicine in reshaping care for persons with movement disorders. *Mov. Disord.* 35, 1897–1902 (2020).
- Tang, Y., Yang, K., Zhao, J., Liang, X. & Wang, J. Evidence of repurposing drugs and identifying contraindications from real world study in Parkinson's disease. ACS Chem. Neurosci. 10, 954–963 (2019).
- Postuma, R. B. et al. MDS clinical diagnostic criteria for Parkinson's disease. Mov. Disord. 30, 1591–1601 (2015).
- Peto, V., Jenkinson, C. & Fitzpatrick, R. PDQ-39: a review of the development, validation and application of a Parkinson's disease quality of life questionnaire and its associated measures. J. Neurol. 245, S10–S14 (1998).
- Chen, K. et al. Evaluation of PDQ-8 and its relationship with PDQ-39 in China: a three-year longitudinal study. *Health Qual. Life Outcomes* 15, 170 (2017).
- Ertan, F. S., Ertan, T., Kiziltan, G. & Uygucgil, H. Reliability and validity of the Geriatric Depression Scale in depression in Parkinson's disease. *J. Neurol. Neurosurg. Psychiatry* 76, 1445–1447 (2005).
- Goetz, C. G. et al. Movement Disorder Society-sponsored revision of the Unified Parkinson's Disease Rating Scale (MDS-UPDRS): scale presentation and clinimetric testing results. *Mov. Disord.* 23, 2129–2170 (2008).
- Tomlinson, C. L. et al. Systematic review of levodopa dose equivalency reporting in Parkinson's disease. *Mov. Disord.* 25, 2649–2653 (2010).
- Kumar, N. et al. Impact of home confinement during COVID-19 pandemic on Parkinson's disease. *Parkinsonism Relat. Disord.* 80, 32–34 (2020).
- Del Prete, E. et al. Prevalence and impact of COVID-19 in Parkinson's disease: evidence from a multi-center survey in Tuscany region. J. Neurol. 268, 1179–1187 (2021).
- 22. Song, J. et al. The changes of exercise pattern and clinical symptoms in patients with Parkinson's disease in the era of COVID-19 pandemic. *Parkinsonism Relat. Disord.* **80**, 148–151 (2020).
- Salari, M. et al. Incidence of anxiety in Parkinson's disease during the coronavirus disease (COVID-19) pandemic. *Mov. Disord.* 35, 1095–1096 (2020).
- 24. Brown, E. G. et al. The effect of the COVID-19 pandemic on people with Parkinson's disease. J. Parkinsons Dis. 10, 1365–1377 (2020).
- Janiri, D. et al. COVID-19 pandemic and psychiatric symptoms: the impact on Parkinson's disease in the elderly. *Front. Psychiatry* 11, 581144 (2020).
- Bot, B. M. et al. The mPower study, Parkinson disease mobile data collected using ResearchKit. Sci Data 3, 160011 (2016).
- Tsang, K. L. et al. Translation and validation of the standard Chinese version of PDQ-39: a quality-of-life measure for patients with Parkinson's disease. *Mov. Disord.* 17, 1036–1040 (2002).

Acknowledgements

The authors thank the patients who participated in this study. This work was supported by the Shanghai Municipal Science and Technology Major Projects (2018SHZDZX01 & ZJLab, 2017SHZDZX01), grants (91949118, 81771372, and 82001201) from the National Natural Science Foundation of China, Shanghai Sailing Program (20YF1404000). Prof. BR Bloem was supported by a center of excellence grant of the Parkinson's Foundation.

Author contributions

J.W., J.J.W., and Y.L.T. conceived and designed the study. J.W. and J.J.W. performed PD diagnostic. Y.L.T., X.N.L., B.S., and J.W. performed statistical and epidemiological analyses. Y.L.T., X.N.L., B.S., and J.W. carried out data analyses and interpretation. Y.L.T., X.N.L., and B.S. wrote the first draft of the manuscript. B.R.B., J.J.W., J.W.W., and J.W. edited and revised the manuscript. J.W. and J.J.W. provided the resources for the survey.

ARTICLE

All authors approved the final version of the manuscript. All authors had full access to all the data used in this study and had final responsibility for the decision to submit for publication.

Competing interests

The authors declare no competing interests.

Additional information

Supplementary information The online version contains supplementary material available at https://doi.org/10.1038/s43856-023-00246-4.

Correspondence and requests for materials should be addressed to Jianjun Wu or Jian Wang.

Peer review information *Communications Medicine* thanks the anonymous reviewers for their contribution to the peer review of this work.

Reprints and permission information is available at http://www.nature.com/reprints

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this license, visit http://creativecommons.org/ licenses/by/4.0/.

© The Author(s) 2023